Author’s response to reviews

Title: Predicting Diabetes Clinical Outcomes Using Longitudinal Risk Factor Trajectories

Authors:

Gyorgy Simon (simo0342@umn.edu)

Kevin Peterson (peter223@umn.edu)

Regina Castro (castro.regina@mayo.edu)

Michael Steinbach (stei0062@umn.edu)

Vipin Kumar (kumar001@umn.edu)

Pedro Caraballo (caraballo.pedro@mayo.edu)

Version: 4  Date: 06 Dec 2019

Author’s response to reviews:

As per the Editor's request, we made the following changes:

1) Moved the Figure Title to a section after the References

2) Created a table of abbreviations (After the References)

3) Updated the Data Availability section to reflect that we cannot share patients' data

4) We stated that the consent was written or electronic

5) We stated that the funding agencies did not contribute to or influence the study design, the collection, analysis or interpretation of the data

6) We added VP's contribution. Now each author has a specific stated contribution.

7) We numbered the references.

8) We made sure that the tables are cited sequentially (Table 3 was cited before Table 2 - &gt; this is resolved.)

9) We reordered the statements in the Declarations as instructed.

Thank you for your patience and help.